Author's response to reviews

Title: Anthropometric, biochemical and clinical assessment of malnutrition in Malaysian patients with advanced cirrhosis

Authors:

Mei-Ling Sharon Tai (sharon@ummc.edu.my)
Khean-lee Goh (klgoh56@streamyx.com)
Siti Hawa Mohd Taib (msitihawa@yahoo.com)
Sanjay Rampal (rampal.s@gmail.com)
Sanjiv Mahadeva (sanjiv@ummc.edu.my)

Version: 4 Date: 19 June 2010

Author's response to reviews:

Dear Sir

Thank you for requesting changes to the manuscript as follows:
1. Major - Full affiliation details have been included in this revised version
2. Minor
   - tracking changes have been turned off
   - author list capitalisation has been amended
   - keywords line has been removed
   - vertical grid lines of tables have been removed

Yours sincerely

Dr Sanjiv Mahadeva